



University of Groningen

Measuring the EU value added embodied in EU foreign exports by consolidating 27 national supply and use tables for 2000-2007

Bouwmeester, Maaike C.; Oosterhaven, Jan; Rueda-Cantuche, José M.

IMPORTANT NOTE: You are advised to consult the publisher's version (publisher's PDF) if you wish to cite from it. Please check the document version below.

Document Version Publisher's PDF, also known as Version of record

Publication date: 2014

Link to publication in University of Groningen/UMCG research database

Citation for published version (APA): Bouwmeester, M. C., Oosterhaven, J., & Rueda-Cantuche, J. M. (2014). Measuring the EU value added embodied in EU foreign exports by consolidating 27 national supply and use tables for 2000-2007. (SOM Research Reports; Vol. 14004-EEF). Groningen: University of Groningen, SOM research school.

Copyright

Other than for strictly personal use, it is not permitted to download or to forward/distribute the text or part of it without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license (like Creative Commons).

Take-down policy If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

Downloaded from the University of Groningen/UMCG research database (Pure): http://www.rug.nl/research/portal. For technical reasons the number of authors shown on this cover page is limited to 10 maximum.



university of groningen

 faculty of economics and business

14004-EEF

Measuring the EU value added embodied in EU foreign exports by consolidating 27 national supply and use tables for 2000-2007

Maaike C. Bouwmeester Jan Oosterhaven José M. Rueda-Cantuche



 faculty of economics and business

SOM is the research institute of the Faculty of Economics & Business at the University of Groningen. SOM has six programmes:

- Economics, Econometrics and Finance
- Global Economics & Management
- Human Resource Management & Organizational Behaviour
- Innovation & Organization
- Marketing
- Operations Management & Operations Research

Research Institute SOM Faculty of Economics & Business University of Groningen

Visiting address: Nettelbosje 2 9747 AE Groningen The Netherlands

Postal address: P.O. Box 800 9700 AV Groningen The Netherlands

T +31 50 363 7068/3815

www.rug.nl/feb/research

university of groningen

Measuring the EU value added embodied in EU foreign exports by consolidating 27 national supply and use tables for 2000-2007

Maaike C. Bouwmeester University of Groningen

Jan Oosterhaven University of Groningen j.oosterhaven@rug.nl

José M. Rueda-Cantuche European Commission, Pablo de Olavide University

Measuring the EU value added embodied in EU foreign exports by consolidating 27 national supply and use tables for 2000-2007[‡]

Maaike C. Bouwmeester^a, Jan Oosterhaven^{a*}, José M. Rueda-Cantuche^b

^a University of Groningen, Faculty of Economics and Business, Postbus 800, NL 9700 AV Groningen, The Netherlands ^b European Commission, Institute for Prospective Technological Studies, Inca Garcilaso 3, E 41092 Seville, Spain

Abstract

This paper develops a method to consolidate national supply-use tables (SUTs) into a single supra-regional SUT. The method deals with mirror trade statistics problems, such as the different valuation of imports and exports, and it corrects for the double-counting of re-exports. To test the contribution of the various construction steps, the paper decomposes the EU value added that is embodied in the EU exports to third countries into seven components. When the national SUTs for the period 2000-2007 are used, neglecting intra-EU spillover and feedback effects between the 27 EU-members results in an underestimation of the embodied value added of 12-15%. Not consolidating the national tables leads to a further underestimation of 11-16%. Both types of errors are substantial. With these underestimations removed, the exports to third countries still only explain around 11% of the EU27 GDP.

Keywords: supply-use tables, mirror trade problems, re-exports, intra-EU spillovers effects, value added in trade

^{*} An extended version of this working paper is forthcoming in *Economic Systems Research* under a different title. The extended version includes a discussion of the differences between consolidating national SUTs and aggregating a partly non-survey international SUT. It also includes a decomposition of embodied CO₂ emissions, as additional test of our method, and a comparison of the CO₂ results with the value added results discussed in this working paper.

^{*} Corresponding author. Faculty of Economics and Business, University of Groningen, Postbus 800, NL 9700 AV Groningen, The Netherlands. Phone +31 50 363 3728; fax +31 50 363 7337; email address <u>j.oosterhaven@rug.nl</u>.

1. Introduction

The European Union (EU) acknowledges openness to trade as a core component of the renewed strategy for economic growth and job creation (Council of the EU 2009, European Commission 2010a). Increasing the competitiveness of the EU in global markets is considered to be of key importance to a trade policy that fits EU's 2020 strategy (European Commission 2010b). The size and importance of the external market are emphasized by the European Commission (2010b, p. 4): 'Our economy is the largest in the world. It is also the biggest exporter. Our firms exported \in 1.6 trillion of goods and services in 2009, which is about 13 % of our GDP.'

Although the volume of EU's external trade is sizeable, assessing its importance as percentage of GDP produces an overstatement. The comparison goes awry by relating trade flows, which are gross flows, to the value added of an economy, which is a net flow (Brahmbhatt, 1998). Trade flows include flows of intermediate goods and services, whereas GDP explicitly excludes these. Intermediate goods and services account for as much as 61% to 63% of EU external imports over the years 2000 to 2007.¹ However, in policy analysis the use of trade/GDP ratios as measure of openness to trade (Roca Zamora, 2009) or as market integration indicator (Ilkovic, 2007) is still commonplace.

Recent literature has focused on measuring trade as a net flow by computing the value added content of trade (Belke and Wang 2006; Daudin *et al.* 2011).² Johnson and Noguera (2012) show the value added content of exports is about 73% of gross exports with substantial variation between countries. The variation in these value added content to gross trade (VAX) ratios is largely attributed to the variation in sector composition of exports. Countries that primarily export manufacturing products tend to have lower VAX ratios, which should have consequences for their score on openness to trade. We will not calculate VAX ratios, but directly concentrate on the methodology to measure the value added content of trade at the supra-regional level.

The increasing importance of decision making at this level, like the European Union and the Euro-zone, underlines the need for model building at that same level. To enable such model building, consistent time series of consolidated national accounts are a *conditio sine qua non*. These times series need to be derived from supply and use tables (SUTs) in order to increase their empirical reliability, to secure their internal consistency, and to estimate interindustry models, such as input-output and general equilibrium models. The required set of supply, use and input-output tables covering a series of years and all EU Member States has been incomplete until recently.

¹ Derived from the input-output tables used in this paper, available via (last accessed 06-03-2012): <u>http://epp.eurostat.ec.europa.eu/portal/page/portal/esa95_supply_use_input_tables/data/workbook</u>

 ^S/₂ Recent working papers are: Koopman *et al*, 2008; Koopman *et al*, 2010, Foster *et al*, 2011.

Within this context, the first contribution of this paper is to develop a method for the consolidation of national SUTs into a single consolidated supra-regional SUT. The method will include a solution to a series of methodological problems, such as the different valuation of national domestic use at basic prices and of imports at c.i.f. prices, and the double-counting of re-exports.³ The second contribution of this paper is an evaluation of the extent to which the use of existing national SUTs instead of the new consolidated EU27 SUT will misestimate the EU27 value added embodied in the EU27 exports to third countries.

The description of the consolidation process starts from a harmonized set of national SUTs in Section 3.⁴ Before that, Section 2 discusses related research and earlier attempts to construct supra-regional SUTs and supra-regional input-output tables (IOTs), in order to indicate what our methodology adds to the existing literature. A crucial element in our method is the repricing of the sum of the national import tables from the basic prices of the importing country into the basic prices of the exporting country (i.e. exclusive of all taxes, subsidies, and trade and transport margins). If this is not done properly, impact analyses of policy measures and scenario analyses will attribute all kinds of indirect effects to the wrong industries (i.e. to commodity producing industries instead of to the government and to trade and transport industries). In addition, double-counting of re-exports and statistical discrepancies also play a role in miscalculating indirect effects, particularly with respect to intra-European trade. At the end of Section 3 it is indicated how the consolidated SUTs are transformed into consolidated IOTs, which are needed to formulate the input-output models used the next section.

Section 4 shows how the correct estimation of the impact of EU27 exports to third countries on EU27 value added may be decomposed into: domestic impacts, first round intra-EU spillover effects, higher round intra-EU spillover and feedback effects, and four types of measurement errors of these effects that occur when national IOTs are used instead of the consolidated EU27 IOT. Section 5 then shows the actual empirical size of these three impacts and the four measurement errors for the period 2000-2007, both at the level of the 59 homogenous EU-branches and for the EU27 as a whole. It appears that not including the intra-EU spillovers and feedbacks leads to an aggregate underestimation of the embodied EU27 value added of 12-15%, while the four measurement errors together lead to an additional aggregate underestimation of the intra-EU spillover and feedback effects of 11-16% over this period. Even with these underestimations removed, EU27 exports to third countries still explain only a little more than 11% of EU27 gross value added or GDP. Section 6 concludes, inter alia, that our results show that it may be more important to strengthen the EU's internal competitiveness than the EU's external competitiveness.

³ The result of the method is a set of annual consolidated supply, use and input-output tables for the EU27 for 2000-2007, published on the Eurostat website last May 2011. Technical documentation related to the dataset (Eurostat, 2011) is available at: <u>http://epp.eurostat.ec.europa.eu/portal/</u>pls/portal.wwpob_page.show? docname=2530266.pdf

⁴ To derive that set for the EU27, lacking SUT data needed to be estimated. See Rueda-Cantuche et al. (2009) for an early account, and see Eurostat (2011) for the final details.

2. Data sources and literature overview

2.1 Data sources

For a correct consolidation of national SUTs into a single supra-regional SUT and IOT, a full set of national SUTs at basic prices is required.⁵ These national SUTs need to include a distinction between intra-European and extra-European exports and imports. Such a database did not exist until recently. The annual national SUT and IOT database, generated jointly by Eurostat and the European Commission's Joint Research Centre (JRC) finds its first use in this paper. This database allows us to estimate both a consolidated EU27 and a consolidated Euro-area time series of SUTs and IOTs for the period 2000-2007. The authors participated actively and collaborated with Eurostat and the JRC, not only in estimating the missing national SUTs and IOTs, but also in developing the consolidation method described in this paper. The final database and the method of consolidation have been endorsed by a technical group of European National Statistical Offices and the European Central Bank for future internal use by Eurostat (see further Eurostat, 2011).⁶

2.2 Construction of IOTs

Input-output tables are usually derived from supply and use tables. The Supply table consists of a matrix of products produced by industries, plus additional rows comprising: imports, distribution margins (trade and transport), and taxes less subsidies on products. The Use table consists of a matrix with the use of intermediate products by domestic industries, and the use of consumption and investment products by households, governments and investing industries. Besides, there are additional columns with changes in stocks and exports, and additional rows with the various components of gross value added, such as labour costs, capital use, net taxes on production and net operating surplus. Such a rectangular SUT, with, for example, *m* industries and *n* products, represents the most appropriate framework for balancing the supply and demand of products, and to compile the Gross Domestic Product (GDP), because it is not based on analytical assumptions, but on statistical data.

Symmetric IOTs can be derived from SUTs by means of behavioural assumptions. Their dimension can be either product-by-product or industry-by-industry.⁷ The fact that IOTs are

⁶ The IOT tables constructed with the method described in this paper can be downloaded from: <u>http://epp.eurostat.ec.europa.eu/portal/page/portal/esa95_supply_use_input_tables/data/database</u>. The relevant metadata can be found at:

http://epp.eurostat.ec.europa.eu/cache/ITY_SDDS/en/naio_esms.htm

⁵ For a detailed definition of basic prices, see Eurostat (2008, p. 163). Basically, basic prices refer to the valuation of goods and services before they are conveyed to markets.

⁷ Rueda-Cantuche (2011) discusses the pros and cons of the choice of dimension of IOTs. Briefly speaking, product-by-product tables are more homogeneous in their description of the transactions. They are the most commonly used tables in input-output analysis. However, they require labour

square is crucial for input-output modelling, as that requires a homogenous representation of the relationships between either products or industries. Productivity, energy and environmental analyses are well-known examples of impact studies for which symmetric IOTs need to be constructed. The same holds for our own decomposition of the error of using national IOT to estimate the EU value added that is embodied in EU exports to third countries.

2.3 Construction of consolidated IOTs

There are essentially two approaches to construct an IOT for a group of countries. First, one may use a variant of what are known as *non-survey* methods. This type of method is generally used for a top-down estimation of a regional IOT from a national IOT (see Miller & Blair, 2009, for an overview). The subset of methods that uses another region's IOT or the national IOT to estimate an unknown regional IOT (e.g. Hewings, 1977) is suited to estimate unknown supranational IOTs. This approach requires that, for the target supra-national IOT, total output, value added, total use, and imports and exports are given by industry, along with at least the totals, but preferably the full columns of the various domestic final demand categories. These totals are then used to define the row and column constraints of the unknown inner part of the target IOT, while the IOT of another country is used as the base for the application of the RAS biproportional adaptation method (see Bacharach, 1970, for an extensive treatment, and see Lenzen *et al.* 2004, and Eder *et al*, 2006 for applications to the Rest of the World and the EU).⁸

Here, however, we apply a method that uses mainly *survey* data to estimate the target supranational IOT. For such a method, there are essentially two alternative routes. The route followed in this paper involves the direct aggregation of the national IOTs and the intra-group imports and exports of the countries concerned. Remarkably, this route has hardly been chosen before, even though national IOTs are available for many countries.⁹ In that sense this paper presents a new and more direct approach. The second route involves the estimation of a full international IOT, which is then aggregated to the required supra-regional level. This approach requires much more work, but also generates more information. Eurostat has opted for the less resource intensive, less problematic and more direct route.

intensive compilation and are based on analytical assumptions that take final results away from observed market transactions. Hence, the integration with other statistical sources is more difficult. Industry-by-industry tables, on the other hand, are much closer to statistical sources; they allow for an easier comparability with other statistical databases and they are less labour intensive to compile.

⁸ Recently Minguez *et al.* (2009) developed a cell-correction to the RAS method (CRAS), which uses not a single IOT but multiple IOTs to improve the estimate of the target IOT (see Oosterhaven & Escobedo, 2011, for the first cross-regional application).

⁹ See Ungar & Heuschling (1994) for the construction of an EU12 IOT for 1991 based on 1985 national tables, Brockmeier (1997) for an EU12 IOT for 1992, Van Leeuwen & Verhoog (1999) for an EU15 IOT for 1993 and Van Leeuwen (2002) for an EU15 IOT for 1995. The documentation of the construction of these tables is, however, rather scanty. More recently, Rueda-Cantuche et al. (2009) constructed an EU27 IOT for 2000.

All earlier work relates to the direct *bottom-up* construction of a supra-regional IOT from national IOTs. Nowadays, however, practically all IOTs are constructed from SUTs (Eurostat, 2008; European Commission et al. 2009). Hence, this will most certainly also hold for the future construction of supra-regional IOTs. Only recently, a start has been made with the construction of international IOTs from international SUTs, due to two large EU-financed research projects, namely EXIOPOL (see Tukker *et al.* 2013) and WIOD (see Dietzenbacher *et al.* 2013). In that sense, this paper also presents a new approach, as we derive a consolidated group SUT directly from national SUTs.

The first international IOTs were constructed for the much smaller old EU for 1959, 1965, 1970 and 1975 (Schilderinck, 1984). The harmonized IOTs of the individual countries for those years already had a split up of the matrix with foreign imports, into a matrix with intra-EU imports and one with extra-EU imports, as well as a comparable split-up of the column with foreign exports. The essence of the construction method consisted of further splitting up the national intra-EU import matrices by country of origin. This was done by means of row-wise uniform national import ratios, which were derived from international import statistics. The row totals of full matrix with these bilateral import flows implicitly represent intra-EU exports of the industry and EU-country at hand. The difference with the actual intra-EU exports was assigned to a column with so-called "expenditure balances".

This procedure is statistically clear, but economically unsatisfactory, as the off-diagonal submatrices with the bilateral trade flows were measured in ex customs' import prices, whereas the domestic transaction submatrices on the diagonal of the IO checkerboard were measured in producer prices. Moreover, the intra-EU export matrices were estimated by means of the import data of the receiving country, without taking the size and composition of their own intra-EU export data in to account. Finally, double counting intra-EU re-exports was not corrected for.

Most likely for these reasons, the resulting "expenditure balances" were as large as 11.5% for manufacturing, 21.5% for fuel and power and even 40% for agriculture (see Oosterhaven, 1995). The current WIOD project partly follows the same procedure in constructing a time series of intercountry SUTs for 1995-2009 with 40 countries, but it combines its equivalent of Schilderinck's "expenditure balances" with the exports to their small Rest of the World (RoW). Given the nature of this procedure and the smallness of their RoW, some of these export flows may turn out to be negative for some products.

Van der Linden and Oosterhaven (1995) re-estimated Schilderinck's intercountry EU-IOTs for 1965-1975, and elongated the series for 1975-1985. They rearranged the re-export flows to avoid double counting, and used RAS to re-price and balance the bilateral intra-EU import tables with the rescaled intra-EU export columns of the countries at hand. The difference between the original intra-EU exports and its rescaled equivalent was put in a "rescaling column" that amounted to only 1% of the original intra-EU exports value.

The second major effort to construct intercountry IOTs is the Asian Input-Output Table (AIOT) project (Meng *et al.*, 2013). The AIOTs combine nine Asian countries and the USA, and are constructed with five year intervals for 1985-2000. The harmonization of the ten national IOTs is the most complicated part of the construction of the AIOTs. The rest of the construction, in essence, follows the EU-method, while incorporating additional information from two extensive surveys. Opposed to the EU-approach, however, the AIOT-method does not use RAS to re-price the import matrices, but does so directly. Moreover, it rescales the bilateral imports to match the bilateral exports total, and combines the rescaling difference with the imports from third, non-AIOT countries, which has the disadvantage of possibly distorting the IO column coefficients.¹⁰

The use of row-wise uniform ratios has been termed the 'proportionality assumption' in recent literature (Trefler and Zhu, 2010). Some evidence shows that international IOTs obtained by applying the proportionality assumption may strongly deviate from results obtained when using survey data. Puzzello (2011) finds that using the proportionality assumption results in countries being a little more intensive in their own factors. Koopman *et al.* (2010) show that it may overestimate the share of intermediate goods in the imports of developed countries, whereas it may underestimate the share of final goods in the exports of many developing countries.

Most recent contributions on the factor content of trade construct an international IOT starting from the GTAP database. The core of the GTAP construction process is the FIT program, which uses entropy methods to update and create the database. It enforces consistency among the IOTs, trade, protection, macro and energy data.¹¹ The GTAP database includes total import matrices and consistent international trade data, but not bilateral trade matrices. Trefler and Zhu (2010) and Johnson and Noguera (2012) both use the proportionality assumption to derive the bilateral trade matrices.

The above discussed international IOT construction procedures all aim to provide the data for the *ideal* international input-output table and model (Isard, 1951). This so-called ideal, however, represents only one member of a whole family of possible multi-national IOTs and models. As regards multi-national SUTs, there also is a whole family of possible accounting schemes with accompanying models (Oosterhaven, 1984).

From this family, the EXIOPOL project and the WIOD project have chosen for the combination of a single intercountry Use table with a set of national Supply tables. From the single intercountry Use table one may derive a full intercountry **A**-matrix with intermediate input coefficients, while from the set of national Supply tables one may derive a single diagonal

¹⁰ Oosterhaven et al. (2008) compare the survey AIOT for 2000 with the outcomes of four simple, less time-consuming, semi-survey methods. All four methods report differences in the intercountry trade flows of 10-50% compared to the semi-survey AIOT for 2000. Using increasingly more information from the export statistics reduces the differences, but not systematically for each and every country. ¹¹ www.gtap.org, last accessed 08-03-2012.

block **S**-matrix with national industry shares in the domestic supply of each product on the diagonal blocks. Together, these two matrices enable the calculation of an intercountry Leontief-inverse, $\mathbf{L} = (\mathbf{I} - \mathbf{S} \mathbf{A})^{-1}$, with all its associated applications.

In summary, the method described in this paper differs from these earlier methods in several ways. First, it is based on a different organization of the data as we use SUTs instead of IOTs, while we only derive a consolidated IOT after constructing the consolidated SUT. Second, we use the available matrices with the intra-EU and extra-EU imports by EU-country, instead of the proportionality assumption at the bilateral country level. Third, our focus on the supranational level allows us to correct the data for re-export flows without making assumptions about the intra-EU geographical distribution of the re-exports. Finally, we re-price the import data of the importing countries into the basic prices of the producing exporting countries.

3. Consolidation of national SUTs into a supra-regional SUT

From the above overview, it follows that the core of the consolidation process requires the redefinition and re-estimation of the member states' intra-EU trade flows, as domestic transactions at the EU27 level. Therefore, a prerequisite for the consolidation procedure is the availability of an intra-EU import use matrix (separated from the extra-EU import matrix) and an intra-EU export column (separated from the extra-EU export column). The method developed in this paper presupposes the availability of 27 harmonized national SUTs, in which this distinction between intra-EU and extra-EU trade has been made.¹²

The construction method for the consolidated EU27 SUT consists of three main steps, and seven sub-steps. First, the national Use tables are aggregated into a *draft* EU27 Use table, which has a structure as shown in Figure 1. It consists of the simple sum of (1) all domestic use tables, (2) all intra-EU import tables, (3) all extra-EU import tables, each including all domestic final use categories, and (4) the simple sum of all intra-EU and extra-EU export columns, and re-exports columns.¹³ Second, in a seven sub-steps procedure, the gray-marked intra-EU trade flows in Figure 1 are merged with the domestic intermediate and final use matrix, right above them, to get the consolidated EU27 Use table. Third, not shown in a figure, the main, domestic industry parts of the national Supply tables are simply aggregated to the domestic industry part of the consolidated EU27 Supply table, while the imports row of the consolidated EU27 supply table equals the simple column-wise aggregation of the matrix with the re-estimated extra-EU import use matrix from Step 2. After this third step, the consolidated EU27 SUT is transformed into the consolidated EU27 input-output table.

¹² See Eurostat (2011) for the details on the construction of the missing intra-EU and extra-EU import matrices.

¹³ The individual country SUTs and this aggregate draft use table have been prepared by Eurostat, the Joint Research Centre's Institute for Prospective Technologies (JRC-IPTS) and the Konstanz University of Applied Sciences (Eurostat, 2011).

Domestic intermediate	Domestic final	1	2		Legend for the o			
use	se use				2: Exports to extr			
		3			3: Re-exports, im			
Intra-EU	Intra-EU final				4 : Re-exports, im			
intermediate			4		5: Re-exports, im			
imports	mports imports				6 : Re-exports, im			
					7: Taxes less subs			
Extra-EU intermediate	Extra-EU final	5	6		8: Taxes less subs			
imports	imports							
	1			1				
TLS	TLS	7	8					

Figure 1: Scheme of the aggregated, draft Use table

columns in Figure 1 a-EU countries

a-EU countries ported from intra-EU, exported to intra-EU ported from intra-EU, exported to extra-EU ported from extra-EU, exported to intra-EU ported from extra-EU, exported to extra-EU sidies on products (TLS) on intra-EU exports sidies on products (TLS) on extra-EU exports

The first and the third step of the overall procedure only require a simple aggregation, which needs no further discussion. The core of the construction method therefore consists of the seven sub-steps needed to derive from Figure 1 the consolidated EU27 use table. The main problem of the seven sub-steps is to balance the intra-EU import table with the information on intra-EU exports in column 1, as each trade flow is reported by two countries, which usually do not match. This problem is known as the *mirror trade statistics puzzle*. There is a series of factors explaining the puzzle, such as time lags between exports and imports, different statistical coverage and response rates, statistical confidentiality, different treatment of revisions, and currency conversion issues (see section 2.17 of Eurostat, 2006).

The economically most important explanation of the puzzle is the structural difference in the prices used: exporting countries usually report their exports in free-on-board (f.o.b.) prices, whereas importing countries usually report their imports in cost-insurance-freight (c.i.f.) prices (see section 2.13 of Eurostat, 2006). The difference between these two prices is made up of international trade and transport margins. In the f.o.b. priced exports, the exported trade and transport services are recorded in the rows pertaining to the service sectors that produce services. In the c.i.f. priced imports, however, the trade and transport margins are included in the value of the various rows with imported goods. In the aggregated draft EU27 use table, exports are actually valued in basic prices. The difference between the exports in basic prices and the imports in c.i.f. prices consists of three valuation layers (see Table 1):

taxes less subsidies levied in the country of export, trade and transport margins within the country of export, and the international trade and transport margins discussed above.

Country	
R	Exports by R in basic prices of R
R	+ Valuation layer: taxes and subsidies
R	+ Valuation layer: trade and transport
R	= Exports by R in f.o.b. price of <i>R</i>
International	+ Valuation layer: international trade and transport margins
S	= Imports by S in c.i.f. price of S
S	+ Valuation layer: taxes and subsidies
S	+ Valuation layer: trade and transport
S	Imports by S in purchaser prices of S

Table 1: Valuation layers in international trade

In order to merge the intra-EU import use table with the domestic use table, both need to be valued in the same prices. Since all kind of applications require that employment, value added, and environmental impacts are allocated to the industries that actually produce the products traded, it is necessary to measure all transactions in *basic prices of the country of origin*. The domestic use table is already measured in such prices, in contrast to the draft intra-EU import use table in Figure 1. Only the latter therefore needs to be re-priced into basic price of the exporting country. Fortunately, information on intra-EU exports in basic prices can be used to balance the intra-EU import table. This is the most important step in the seven sub-steps procedure that changes the draft EU-Use table into the consolidated EU27 Use table. Table 2 gives an overview of all seven steps.

Step 1:	Adjust for taxes less subsidies on intra-EU imports (cell 7)
Step 2:	Correct trade flows imported from intra-EU, re-exported to intra-EU (column 3)
Step 3:	Correct trade flows imported from intra-EU, re-exported to extra-EU (column 4)
Step 4:	Correct trade flows imported from extra-EU, re-exported to intra-EU (column 5)
Step 5:	Re-scale all import values to the total of the intra-EU exports (column 1)
Step 6:	Balance the rescaled intra-EU import table with the intra-EU export vector, using GRAS
Step 7:	Aggregate the domestic and balanced intra-EU import table

Step 1: Adjust for taxes less subsidies on intra-EU imports

Since intra-EU imports are valued at c.i.f. prices, they include, among others, the taxes less subsidies on products (TLS) that are associated with these transactions. The first step will be to reallocate (deduct) the TLS from the intra-EU import matrix. The total amount to be reallocated is given by cell 7 of Figure 1, which contains the taxes less subsidies linked to the same transactions, as reported by the exporting country. In order to do this, we proportionally distribute the value in cell 7 of Figure 1 over the values in the row identified by 'TLS' in Figure 1.

In other words, we reallocate the net taxes levied by exporting countries to the domestic industries and the domestic categories of final demand of the importing countries by means of the shares of taxes less subsidies paid by these industries, final consumers and investors. The changes in the TLS row represent an average increase of 0.38% over the period 2000-2007, whereas the changes in intra-EU import columns show an average decrease of 0.25%. At the end of this step, which is summarized in Figure 2, the intra-EU imports are adjusted to c.i.f. prices, net of taxes less subsidies.

The next three steps correct re-exports that are correctly measured from the perspective of the country that assembles these data, but not from an intercountry input-output perspective (see Guo *et al.* 2009). The core problem is to reallocate re-exports to the real country of origin. This is increasingly important, as re-exports grow, especially in countries like Hong Kong, Singapore and the Netherlands (see Mellens *et al.* 2007). Existing databases only provide ad hoc solutions for some of these countries (e.g. Feenstra *et al.* 2005, and Narayanan *et al.* 2012). We propose simple solutions, but solutions that are applicable to all 27 EU countries.

Step 2: Correct for double counting intra-EU trade flows that are re-exported within the EU

An example best clarifies how re-exports within the EU are double counted. Take Italian shoes that are re-exported by Austria to the Czech Republic. Clearly, this transaction will be reported in the Italian use table as an intra-EU export (in column 1 in Figure 1), and in the Czech use table as an intra-EU import of final goods. Therefore, the transaction on re-exports reported by the Austrian tables (column 3 in Figure 1) is redundant. Both the exporting country and the importing country record these trade flows in a correct way, so no adjustment has to be made. Maintaining the re-exports of Austria would result in double counting.¹⁴ Therefore, the values in column 3 are simply deleted, as indicated in Figure 3.

¹⁴ By correcting for the re-exports as described in Step 2, 3 and 4, it is assumed that re-exports are not again re-exported by the "final" importer.



Step 3: Correct for trade flows imported from within the EU and re-exported outside the EU

Next, consider the case in which Austria re-exports the Italian shoes to Switzerland (outside the EU) instead of to the Czech Republic (inside the EU). Because Italy not necessarily knows what Austria is doing with its shoes, in this case, Italy wrongly records this transaction as an intra-EU export, whereas it should have been reported as an extra-EU export. When constructing an inter-country SUT the necessary correction cannot be made at the individual country level, because the origin or destination of the re-exports is not recorded in the individual country SUT. At the level of a consolidated EU Use table, however, this correction can be made due to the differentiation between intra-EU re-exports and extra-EU re-exports. Column 4 of Figure 1 depicts precisely the goods and services that are imported from intra-EU countries and re-exported outside the EU. The values of column 4 are therefore subtracted from column 1 with the intra-EU exports, and added to column 2 with the extra-EU exports as indicated in Figure 3. This correction leads to a considerable average decrease of 7.6% in total intra-EU exports, along with an even larger average increase of 12.4% in total extra-EU exports.

Step 4: Correct for trade flows imported from outside the EU and re-exported within the EU

Next, consider the reverse case of the re-export of Swiss chocolate by Spain to Portugal. Portugal wrongly records this transaction as an intra-EU import, since it does not 'know' that Spain has imported these goods from Switzerland. Spain will report this transaction in column 5 of Figure 1. The correction therefore entails reducing the reported imports from EU countries, and increasing the reported imports from countries outside the EU, as indicated in

Figure 4. This is done by subtracting the values of column 5 proportionally, row-by-row, from the values of the intra-EU imports, and adding the subtracted values to the corresponding cells of the extra-EU import matrix. These changes lead to an average decrease of 11.6% in total intra-EU imports, and an average increase of as much as 17.4% in total extra-EU imports over the period 2000-2007.



Figure 4: Step 4

Figure 5: Step 5

Finally, as regards the re-exports, note that the re-export column 6 is maintained without any change in the consolidated table, as indicated in Figure 5. Both the original exporting country and the final importing country are non-EU countries. This combination is correct and does not require adjustments in the consolidated EU27 Use table.

Step 5: Rescale all intra-EU imports such that their total equals that of the intra-EU exports

After the above tax and the re-export corrections, each of the row sums of the adjusted intra-EU import table has to match the values reported in the adjusted intra-EU export column 1. The actual matching is done in the next step, but to undertake this matching successfully, it is required that the two adjusted overall totals are equal. To achieve this, all cells in the adjusted intra-EU import table, i.e., both intermediate and domestic final demand, are further adjusted by multiplying each cell with the sum of the adjusted intra-EU export values and dividing it by the overall sum of the adjusted intra-EU import table, as indicated in Figure 5.

Theoretically, the rescaling factor reflects the share of trade and transport margins included in imported goods from country r by s that do not balance against the trade and transport margins recorded as exports of r to s. This discrepancy occurs when non-EU carriers deliver part of the trade and transport services. For this reason and also because we do not want to change value added by industry or total final demand by category, the matrix with rescaling differences is added to the matrix with the extra-EU imports, as illustrated in Figure 5.

In practice, however, all factors identified to contribute to the *mirror trade statistics puzzle* may contribute to the difference in the adjusted totals of the intra-EU imports and the intra-EU exports. The actual rescaling factor for the EU tables for 2000-2007 is on average 1.10, with all yearly values within a 0.02 maximum deviation from this value. Thus, before rescaling, the adjusted total intra-EU export value was about 10% higher than the adjusted total intra-EU import value. The compensating change in the extra-EU imports amounted to -15%.

Note that this scaling down of the extra-EU imports does not necessarily represent an improvement, as it might be argued that its adjusted value (+17.4% in Step 4) is measured more correctly than the adjusted value of the extra-EU exports (+12.4% in Step 2). In our consolidation method, however, we give precedence to the belief that the split-up between intra- and extra-EU exports is better measured than the split-up between intra- and extra-EU imports.¹⁵ Also note that one of the consequences of our re-estimation of both extra-EU imports and exports is that the EU external trade balance gets roughly +10 percent points larger than it than it was measured originally.¹⁶

Step 6: Balancing the intra-EU import table with the intra-EU export column using GRAS

Next, the Generalized RAS method of Junius & Oosterhaven (2003) is used to balance the adjusted intra-EU import totals by product, with the adjusted intra-EU exports by product. GRAS is a bi-proportional adjustment method very similar to RAS (Stone, 1961). The difference is that it can deal with negative values in the same fashion as with positive values. RAS and GRAS can be applied to any table for which an initial structure is given (or assumed), and new row and column totals are supplied, provided that the total of the row totals equals the total of the column totals, as secured in Step 5. The method's solution is equivalent to adding minimum information to the table obtained after Step 5, such that it just satisfies the new totals. RAS has been widely used to update input-output tables (see Miller & Blair, 2009), but it can also be used to balance import matrices with given export totals, as done by Van der Linden & Oosterhaven (1995) and Oosterhaven *et al.* (2008).

By using the export values in basic prices of the country of origin as row constraints in the GRAS procedure, the trade and transport margins included in the c.i.f. priced import values are effectively redistributed to the rows with the corresponding services. In this way, the

¹⁵ An additional, much weaker argument is that both changes (+17.4% in Step 4 and -15% in Step 5) compensate for each other. Alternatively, the difference between imports and exports could have been added to the extra-EU exports, as was done in the older EU-method discussed in Section 2. That would, however, lead to an even larger upward correction of the extra-EU exports (+12.4% in Step 2 and +15% in Step 5).

¹⁶ This +10 percent point is the result of the +17.4% and -15% in Step 4 and 5 for extra-EU imports and the +12.4% in Step 2 for extra-EU exports.

balanced intra-EU import use table is implicitly re-priced into basic prices. At the end of this step, the adjusted intra-EU export column 1 of figure 5 has become redundant.

Step 7: Aggregation of the domestic use table and the balanced intra-EU import table

In the last step of the Use table consolidation, the adjusted and balanced intra-EU import table is added to the table with the sum of the unchanged domestic use tables. The resulting consolidated EU27 Use table now only contains one import table, namely the imports from extra-EU countries.

Two final observations are of importance. First, note that the balance of total demand and total supply by product, and the balance of total input and total output by industry, in each of the national SUTs, is maintained in the consolidated SUT, while intra-EU export values and intra-EU import values are merged with the domestic transactions.

Second, note that gross domestic product (GDP) of the EU27 is not altered by the consolidation method. In the procedure, the member state's exports are decreased by the amount of intra-EU exports and their imports are decreased by the amount of intra-EU imports, as these flows are merged with the domestic transactions. As both exports and imports decrease by the same amount, the net values of exports less imports do not change, and the EU27 GPD remains unchanged.

Once the consolidated supply and use tables have been constructed, an industry-by-industry consolidated EU27 input-output table (IOT) is estimated by using the *fixed product sales structure assumption* (see Eurostat, 2008, p. 349 – Model D).¹⁷ We opted for an industry-by-industry IOT due to the fact that value added is predominantly linked to industries rather than to products.

4. Decomposition of the 'true' value added embodied in the EU27 exports

We evaluate the empirical consequences of the different consolidation steps by estimating the EU27 value added embodied in the EU27 exports to third countries (extra-EU exports). This specific test is chosen for two reasons. First, value added impacts are chosen as GDP represents the single most important policy indicator available. Second, extra-EU exports are chosen to weight the implicitly used value added multipliers, as external competitiveness is a major EU policy goal. Besides, it is of direct interest to evaluate the actual importance of extra-EU exports for the EU-value added by industry.

The empirical consequences of the new consolidation method are evaluated in two main steps. First, we estimate the embodied EU27-value added by means of the national IOTs, i.e. by summing the outcomes of 27 national Leontief models. Second, we study how this

¹⁷ Note, however, that the Eurostat consolidated IOTs are of the product-by-product type.

summed estimate changes when the consolidated IOT is used to specify a single Leontief quantity model for the whole of the EU27.

The core variables of the standard input-output (IO) model are: $\mathbf{x} = a$ column vector with total output by industry, $\mathbf{Z} = an$ industry-by-industry matrix with intermediate inputs, and $\mathbf{y} = a$ column vector with total final demand (the sum of consumption, investments, government expenditures and exports) by producing industry. In the standard IO model, supply follows demand, i.e., $\mathbf{x} = \mathbf{Z} \mathbf{i} + \mathbf{y}$ (with \mathbf{i} being a summation vector of ones). Final demand \mathbf{y} is assumed to be determined exogenously, while intermediate demand is determined by total output per industry, i.e., $\mathbf{Z} \mathbf{i} = \mathbf{A} \mathbf{x}$. The matrix \mathbf{A} , with the intermediate input coefficients, is calculated from the IOT data by means of $\mathbf{A} = \mathbf{Z} \hat{\mathbf{x}}^{-1}$ (with $\hat{\mathbf{x}}$ being a diagonal matrix of \mathbf{x}). The well-known solution of the standard IO model then reads: $\mathbf{x} = (\mathbf{I} - \mathbf{A})^{-1} \mathbf{y}$.

In our application of this model we only add **v**', the IOT row with total value added by industry, and let it be determined by total output by industry, i.e., $\mathbf{v}' = \mathbf{c}'\hat{\mathbf{x}}$, where **c**' is a row with value added coefficients, calculated from the IOT data by means of $\mathbf{c}' = \mathbf{v}'\hat{\mathbf{x}}^{-1}$. The test variable in this paper only considers a part of final demand, namely, the EU27 exports to third countries **e**. Adding **c**' and **e** to the standard model allows us to calculate which part of the value added of each industry is embodied in the extra-EU exports of each commodity. Using diagonal matrices of **c** and **e**, this can be done in a very detailed way: $\mathbf{V} = \hat{\mathbf{c}}(\mathbf{I} - \mathbf{A})^{-1}\hat{\mathbf{e}}$.

Summing **V** across its rows gives the value added of the EU-industry at hand that is embodied in the extra-EU exports of all commodities. Summing **V** across its columns gives the value added in all EU-industries that is embodied in the extra-EU exports of the commodity at hand. Here we report only the results for the latter summation, as its outcomes are more relevant for policy formulation, which mostly focuses on the exporting industries.

A first approximation of the EU-wide value added embodied in the extra-EU exports, using only national IOT data, can be made by taking the *EU-sum of the domestic impacts* of each and every Member State's exports to third countries:

$$\mathbf{v}(\mathbf{1a}) = \sum_{r} \mathbf{c}^{r'} \left(\mathbf{I} - \mathbf{A}^{rr} \right)^{-1} \hat{\mathbf{e}}^{r}$$
(1a)

where *r* = 1, ..., 27 (EU-member countries), and

 $\hat{\mathbf{e}}^r$ = diagonal matrix with extra-EU exports by industry of member state *r*, taken from the final demand part of the IOT of member state *r*, i.e. uncorrected for re-exports.

 A^{rr} = industry-by-industry matrix of domestic intermediate input coefficients of country r (this matrix does not change during the consolidation procedure), and

 $\mathbf{c}^{r'}$ = row with the value added coefficients by industry of country *r* (this vector also does not change during the consolidation procedure).

Evidently, (1a) systematically under-estimates the EU-wide impacts, as it ignores each country's spillover effects on the value added of the remaining EU member states, caused by its needs of imports from these member states. This means, for example, that the value added of French car parts' producers that service the German car exports to China is ignored. These *first order intra-EU spillovers*, however, can also be estimated by means of national IOT data, namely as follows:

$$\mathbf{v}(\mathbf{1b}) = \sum_{r} \mathbf{c}^{r'} \mathbf{A}^{er} \left(\mathbf{I} - \mathbf{A}^{rr} \right)^{-1} \hat{\mathbf{e}}^{r}$$
(1b)

where, additionally:

 A^{er} = interindustry matrix of intra-EU intermediate import coefficients of member state *r*, taken from *r*'s national IOT, i.e. uncorrected for re-pricing and balancing these imports with the corresponding exports.

Nevertheless, taking the sum of (1a) and (1b) still systematically underestimates the EU-wide impacts of the extra-EU exports. This second underestimation results from the fact that the *higher order intra-EU spillovers and feedbacks* are not included. This means, for example, that the value added of subpart producers that service the French car part producers that service the German car export to China is not included. When these subpart producers are located in Germany one speaks of intra-EU *feedback* effects of German exports via the rest of the EU27 back to German industries. When the subpart producers are located in the rest of the EU one speaks of *higher order* intra-EU *spillover* effects (see Oosterhaven, 1981; Miller and Blair, 2009). These higher order effects may be estimated, again using only national IOTs, by taking the difference between the total EU-wide impacts, and the sum of (1a) and (1b), as follows:

$$\mathbf{v}(\mathbf{1c}) = \sum_{r} \mathbf{c}^{r'} \left(\mathbf{I} - \mathbf{A}^{rr} - \mathbf{A}^{er} \right)^{-1} \hat{\mathbf{e}}^{r} - \mathbf{v}(\mathbf{1a}) - \mathbf{v}(\mathbf{1b})$$
(1c)

Note that taking the sum of \mathbf{A}^{rr} and \mathbf{A}^{er} in (1c) implies the use of the incorrect assumption that, for example, the French car part products that are exported to the German car industry are produced by means of the technology of the German car part producers.

The sum of (1a), (1b) and (1c) gives an estimate of the *total EU-impacts* of all extra-EU exports when only national IOTs would be available. This total can, of course, also be calculated directly as:

$$\mathbf{v}(1) = \sum_{r} \mathbf{c}^{r'} \left(\mathbf{I} - \mathbf{A}^{rr} - \mathbf{A}^{er} \right)^{-1} \hat{\mathbf{e}}^{r}$$
(1)

but having only the outcomes of (1) does not allow us to study the decomposition of the EUwide impact into domestic impacts (1a), first order intra-EU spillovers (1b), and higher order intra-EU spillovers and feedbacks (1c).

Next follows the question whether the construction of a consolidated SUT, and its transformation into a consolidated IOT, does make a difference to the outcomes of (1). To study this, we separately consider the consequence of the adjustments made to each of the sets of coefficients present in (1).

The first source of error in (1) relates to using c''. These national value added coefficients are used correctly in (1a) to estimate the domestic value added impacts, but they are used in (1b) as a proxy for the weighted average of the value added coefficients of the rest of the EU, while they are used as a proxy for the value added coefficients of the whole of the EU in (1c). Using the true value added coefficients for the EU27 in (1), corrects for these two errors in one go. The related *value added coefficient error* can therefore be calculated as follows:

$$\mathbf{v}(2\mathbf{a}) = \sum_{r} \left(\mathbf{c}^{e} - \mathbf{c}^{r} \right)^{\prime} \left(\mathbf{I} - \mathbf{A}^{rr} - \mathbf{A}^{er} \right)^{-1} \hat{\mathbf{e}}^{r}$$
(2a)

with:

 \mathbf{c}^{e} = vector with EU27 value added coefficients.

Note that this measurement error does not have a systematic bias. We expect that the underestimations caused by countries with small value added coefficients will compensate the overestimations of those with large value added coefficients. On average, the value added coefficient error should be close to zero.

Second, we consider the impact of aggregating the 27 national SUTs into the consolidated EU27 SUT that is directly derived from Figure 1, i.e. without redistributing taxes and reexports, and without re-pricing and balancing the intra-EU import table with the re-estimated intra-EU exports column. After this crude SUT aggregation, a non-corrected EU27 IOT was derived by means of the fixed product sales structure assumption, in precisely the same way as the national IOTs were derived from the national SUTs. The *aggregation error* is calculated as follows:

$$\mathbf{v}(2\mathbf{b}) = \mathbf{c}^{e'} \left(\mathbf{I} - \mathbf{A}_{non}^{ee} \right)^{-1} \hat{\mathbf{e}}_{non}^{e} - \sum_{r} \mathbf{c}^{e'} \left(\mathbf{I} - \mathbf{A}^{rr} - \mathbf{A}^{er} \right)^{-1} \hat{\mathbf{e}}^{r}$$
(2b)

with, additionally:

 \mathbf{A}_{non}^{ee} = interindustry matrix with the non-corrected intra-EU27 input coefficients, i.e., with the industry output weighted average of all $(\mathbf{A}^{rr} + \mathbf{A}^{er})$, which is calculated by post-multiplying the sum of the domestic input coefficients and the intra-EU import coefficients, calculated from Figure 1, with the EU-27 domestic industry shares, calculated from the consolidated supply table, and

 $\hat{\mathbf{e}}_{non}^{e}$ = a diagonal matrix with the non-corrected EU27-exports to third countries, i.e. the simple sum of all $\hat{\mathbf{e}}^{r}$ with its diagonal elements taken directly from column 2 of Figure 1.

Note that this aggregation error also does not have a systematic bias. We expect that the average aggregation error will be close to zero.

Third, the consolidation procedure includes the treatment of the re-exports, detailed in Section 3. Step 3 in that section corrects for intra-EU exports that are actually re-exported to third countries. The non-treatment of these re-exports leads to an underestimation of the impulse vector, i.e. the extra-EU exports, of around 11% on average over the period 2000-2007. This error due to the *underestimation of extra-EU exports* can simply be calculated as follows:

$$\mathbf{v}(2\mathbf{c}) = \mathbf{c}^{e'} \left(\mathbf{I} - \mathbf{A}_{non}^{ee} \right)^{-1} \left(\hat{\mathbf{e}}^{e} - \hat{\mathbf{e}}_{non}^{e} \right)$$
(2c)

where: $\hat{\mathbf{e}}^{e}$ = a diagonal matrix with the corrected extra-EU exports from the consolidated IOT (column 2 in Figure 3).

Finally, we consider the error that occurs if the sum of the national intra-EU import matrices, from Figure 1, would remain unbalanced with the intra-EU exports. This error consists of four sub-errors:

- 1. The information on taxes less subsidies on intra-EU exports (cell 7 of Figure 2) is used to downscale the intra-EU import values by an average of 0.25% to get c.i.f. prices net of the TLS on products.
- 2. The intra-EU imports are further downscaled by an average of 11.6% to correct for intra-EU imports that are actually re-exported outside the EU (column 5 of Figure 4).
- 3. These downscaled intra-EU imports are scaled up by an average of 9.6% to match the total of the re-estimated intra-EU exports (see Figure 5).
- 4. Finally, these imports are balanced with the intra-EU exports by product, by means of GRAS, which leads to an scaling up of the rows with trade and transport margins and scaling down of the remaining rows, in order to arrive at the basic prices of the exporting country (see Step 6).

The combined *non-balancing error* can be calculated as follows:

$$\mathbf{v}(2\mathbf{d}) = \mathbf{c}^{e'} \left(\mathbf{I} - \mathbf{A}^{ee}\right)^{-1} \mathbf{\hat{e}}^{e} - \mathbf{c}^{e'} \left(\mathbf{I} - \mathbf{A}^{ee}_{non}\right)^{-1} \mathbf{\hat{e}}^{e}$$
(2d)

where, additionally:

 A^{ee} = the "true" interindustry matrix with the intra-EU intermediate input coefficients being derived from the consolidated EU27 input-output table.

The four corrections of the intra-EU import use table, summarized above, without doubt represent a major improvement compared to the crude consolidation of national SUTs. It's aggregate impact, however, will be relatively small, as the four sub-errors largely compensate each other. We therefore expect that its main impact will be found at the industry/ commodity level.

Obviously, the overall error of not having a correctly consolidated SUT, when estimating EUwide value added impacts, is the total of (2a)-(2d). This total error can also be calculated directly as:

$$\mathbf{v}(2) = \mathbf{c}^{e'} \left(\mathbf{I} - \mathbf{A}^{ee} \right)^{-1} \hat{\mathbf{e}}^{e} - \mathbf{v}(1)$$
(2)

but with only the results of (2) it would not be possible to study the decomposition of the errors that is presented in (2a)-(2d). The latter is important, amongst others, because the aggregation error has a different nature than the other errors. The other errors represent errors that are due to *not* consolidating the national SUTs correctly, whereas the aggregation error is an unavoidable error of the consolidation procedure itself. The ideal would be to use a full 27x27 intercountry EU27 SUT, but that SUT would have considerable errors in its intra-EU trade matrices, as a correct re-allocation of the re-export column to the original exporting countries is hardly possible, and because then we would have to use the proportionality assumption to allocate intra-EU trade to the industries and final demand categories of destination. All these errors, which are avoided by the consolidation, may well outweigh the aggregation error.

5. Decomposition of value added embodied in the EU27 exports for 2000-2007

In this section we discuss whether our theoretical expectations about the size of the different impacts (1a)-(1c), and the different measurement errors (2a)-(2d) correspond with the actual empirical outcomes for 2000-2007. To present our results at a manageable level, we only show the exported commodities that have an average impact on the value added of the

whole EU27 economy equal or greater than 1% during the period 2000-2007. This resulted in a selection of 31 commodities out of 59.

The bottom row of Table 3 shows that, on average, the value added embodied in extra-EU exports is underestimated by "about" 27% (100% – 72.9%) if we use the 27 national IOTs rather than the consolidated EU27 IOT. The wording "about" is added, as an unknown aggregation error is related to using a consolidated IOT for the EU27 instead of a full intercountry IOT with harmonized and balanced bilateral trade matrices between all 27 individual EU member states. The actual aggregation error of 0.8% of column *e* that we do make is an unreliable indication of this "true" aggregation error, as it relates to aggregating 27 unadjusted national IOTs instead of aggregating an adjusted full intercountry 27x27 IOT.

Table 3: Decomposition o	EU27 value added embodied in o	extra-EU exports, average for 2000-2007
--------------------------	--------------------------------	---

		Using only national IO models			Errors of not using the consolidated EU27					1
Selected group of commodities with an impact larger than 1%		Domestic impacts	First order Intra-EU spillovers	Higher order intra-EU spillovers and feedback	Due to factor coefficients	Due to aggregation	Due to extra-EU exports	Due to intra-EU imports	Total of non- consolidation	Contribution of extra-EU exports to EU27 value added
		(a)	(b)	(c)	(d)	(e)	(f)	(g)	(h) = (d) + (e) + (f) + (f) + (g)	(i) = (a) + (b) + (c) + (h)
		v1a	v1b	v1c	v2a	v2b	v2c	v2d	v2	v1+v2
23	Machinery and equipment n.e.c.	73.7%	5.9%	8.2%	-0.9%	1.6%	11.8%	-0.2%	12.2%	10.4%
18	Chemicals, chemical products and man-made fibres	62.8%	7.3%	9.7%	-0.2%	1.5%	18.9%	0.1%	20.2%	9.7%
28	Motor vehicles, trailers and semi-trailers	62.6%	7.4%	13.4%	-2.3%	2.4%	16.3%	0.3%	16.6%	9.1%
51	Other business services	84.9%	2.4%	2.6%	6.1%	-0.9%	4.7%	0.3%	10.1%	6.3%
36	Wholesale trade and commission trade services, except of mo	87.9%	3.4%	4.0%	3.4%	-0.1%	1.8%	-0.4%	4.7%	5.8%
9	Food products and beverages	73.0%	5.8%	7.9%	2.6%	-1.0%	11.5%	0.2%	13.3%	4.0%
21	Basic metals	64.3%	7.2%	12.1%	-0.2%	1.7%	14.8%	0.0%	16.4%	3.5%
25	Electrical machinery and apparatus n.e.c.	69.1%	6.3%	9.1%	0.1%	1.0%	15.5%	-1.1%	15.5%	3.4%
29	Other transport equipment	69.7%	6.2%	9.5%	0.3%	0.5%	16.0%	-2.2%	14.6%	3.2%
40	Water transport services	72.6%	9.2%	9.3%	-1.5%	6.0%	4.0%	0.4%	8.9%	2.7%
27	Medical, precision and optical instruments, watches and clock	72.1%	5.0%	6.4%	-1.6%	2.3%	16.6%	-0.9%	16.4%	2.7%
26	Radio, television and communication equipment and apparatu	63.0%	6.8%	9.2%	-2.0%	4.2%	22.9%	-4.1%	21.0%	2.7%
22	Fabricated metal products, except machinery and equipment	76.6%	5.5%	8.5%	-0.5%	1.0%	8.7%	0.3%	9.5%	2.2%
19	Rubber and plastic products	68.2%	7.0%	10.2%	0.3%	1.1%	13.5%	-0.2%	14.6%	2.1%
44	Financial intermediation services, except insurance and pensi	77.5%	4.2%	6.9%	17.9%	-13.5%	6.9%	0.1%	11.4%	1.9%
15	Pulp, paper and paper products	70.4%	6.6%	9.4%	-1.1%	2.4%	12.3%	0.0%	13.6%	1.8%
37	Retail trade services, except of motor vehicles and motorcycl	86.3%	1.6%	2.0%	7.3%	0.9%	1.5%	0.5%	10.2%	1.7%
30	Furniture; other manufactured goods n.e.c.	68.9%	5.9%	8.4%	2.7%	-1.5%	15.4%	0.2%	16.8%	1.7%
42	Supporting and auxiliary transport services; travel agency serv	84.5%	4.2%	5.1%	0.7%	-0.1%	4.3%	1.3%	6.2%	1.5%
35	Trade, maintenance and repair services of motor vehicles and	89.9%	3.1%	4.7%	-3.9%	4.3%	1.2%	0.7%	2.3%	1.5%
49	Computer and related services	86.5%	3.8%	3.4%	1.7%	1.1%	2.9%	0.7%	6.4%	1.5%
11	Textiles	66.5%	7.4%	10.3%	1.1%	0.1%	14.8%	-0.1%	15.8%	1.5%
41	Air transport services	74.4%	5.6%	7.3%	0.4%	1.4%	9.5%	1.3%	12.7%	1.4%
17	Coke, refined petroleum products and nuclear fuels	55.4%	16.3%	12.7%	4.2%	2.9%	11.9%	-3.5%	15.5%	1.4%
46	Services auxiliary to financial intermediation	84.4%	1.9%	2.2%	13.5%	-5.3%	3.3%	-0.1%	11.5%	1.4%
50	Research and development services	82.8%	3.7%	4.4%	1.4%	2.4%	5.6%	-0.4%	9.1%	1.3%
20	Other non-metallic mineral products	81.1%	5.1%	6.3%	0.7%	-0.1%	7.2%	-0.2%	7.6%	1.3%
39	Land transport; transport via pipeline services	87.0%	3.3%	3.8%	0.1%	2.2%	4.3%	-0.6%	6.0%	1.3%
24	Office machinery and computers	43.6%	7.3%	10.3%	3.5%	1.0%	40.4%	-6.0%	38.8%	1.2%
16	Printed matter and recorded media	75.6%	6.5%	6.6%	3.0%	2.6%	5.0%	0.7%	11.2%	1.1%
1	Products of agriculture, hunting and related services	81.1%	3.6%	5.0%	2.2%	-1.0%	8.9%	0.2%	10.3%	1.0%
	Selected total (billion €)	707.1	56.4	78.0	11.4	8.9	113.5	-2.9	130.9	972.3
	Total (billion €)	769.1	59.6	82.2	17.3	7.9	122.0	-3.1	144.2	1,055.1
	Percentage of EU27 value added embodied in exports	72.9%	5.7%	7.8%	1.6%	0.8%	11.6%	-0.3%	13.7%	100.0%
	NOTE: 1 billion = 1000 Mio Euro									

The overall underestimation of 27% consists of two main components: 13.5% point (5.7+7.8) of it is due to the neglect of intra-European spillover and feedback effects (v1), and 13.7% point is due to not consolidating the 27 national IOTs (v2). The highest two lines in Figure 6 illustrate the evolution of these two magnitudes during the whole period of 2000-2007. The error of neglecting intra-EU spillovers and feedbacks varies between 12% and almost 15%, while the error of non-consolidation varies between a little over 11% to almost 16%.

The overall error of 27% of course varies by exported commodity (see column *a* of Table 3). It is largest for office machinery and computers (industry 24, around 56%), followed by a group of eight manufactured goods that all have errors larger than 30%. The overall error is smallest for trade, maintenance and repairing services of motor vehicles, etc. (industry 35, around 10%), followed by a group of nine service industries, agriculture and other non-metallic mineral products that all have overall errors smaller than 20%.



Figure 6: Errors made in estimating the value added embodied in EU27 exports

The size of the separate errors that occur when the consolidated IOT is not used also fluctuates over time as shown in Figure 7. We identified four sources of errors of using only national IOTs to estimate the embodied value added of the extra-EU exports: errors in value added coefficients; errors from aggregation; errors from the non-correction of extra-EU exports; and errors from the non-balancing of intra-EU imports (rescaling and product-wise balancing with extra-EU exports).

The error of using national value added coefficients (v2a) is rather small, as expected, while it becomes practically irrelevant at the aggregate level of the EU27 for the latter years studied. The convergence of the national value added coefficients towards the average EU value

added coefficients suggests that the national production structures within the EU27 have been converging. The especially strong fall from 2003 to 2004 might be due to the accession of the 10 newest EU member states in 2004 and the subsequent application of the European System of Accounts in these countries, particularly with regard to the estimation of value added at basic prices and GDP. The fall in the total non-consolidation error from 2003 to 2004 in Figure 6 is, in fact, mainly due to the reduction in the error of using national value added coefficients shown in Figure 7.



Figure 7: Consolidation errors made in estimating value added embodied in EU27 exports

In line with our expectation of an unbiased aggregation error (v2b), the actual error turns out to be very small (0.8%, see Table 3, column *e*). The largest errors are found for financial services (industry 44) and services auxiliary to financial intermediation (industry 46). They partially compensate the even larger errors of national value added coefficients for the same industries. The large errors in these financial industries most likely are related to methodological differences in the treatment of FISIM¹⁹ services among countries.

The error due to the underestimation of extra-EU exports (v2c) is by far the most significant, being responsible for around 11.6% of the total "true" impact (see Table 3, column f). It also grew persistently throughout the period 2000-2007 (see Figure 7). Its large size can partly be ascribed to the fact that the extra-EU exports constitute the driver of the Leontief quantity model used in all equations of Section 4. The problem of re-exports is particularly heavy for office machinery and computers (40%), followed by a group of thirteen manufactured goods

¹⁹ Financial Intermediation Services Indirectly Measured

that all have errors between 10% and 20%. Not surprisingly, services all have relatively small re-exporting allocation errors.

As discussed in Section 4, the mutually compensating sub-errors of non-balancing the intra-EU imports with the intra-EU exports (v2d) do not result to a large aggregate error, while it is also quite stable over the period 2000-2007. The largest errors are overestimations, and are found for office machinery and computers (industry 24); radio, television and communication equipment (industry 26); and coke and refined petroleum products (industry 17) (see Table 3, column *g*).

Finally, and maybe most importantly, note that the percentage structure of the last column of Table 3 is not related to the consolidation process, instead it gives an indication of the commodity mix information that this type of application of a consolidated SUT and IOT may generate. Remarkable are the large contributions of the exports of machinery and equipment (10.4%), chemicals etc. (9.7%) and motor vehicles etc. (9.1%) to the total value added embodied in extra-EU exports. This total impact, of a little more than 1,000 billion euro, gives an indication of the relatively small importance of the exports to third countries for the value added of the EU27, as this amount represents only 11.2% of European GDP. Obviously, domestic EU27 consumption, investment and government expenditures are responsible for the remaining 88.8% of European GDP.

6. Conclusion and discussion

In this paper we developed a method to consolidate a series of national supply and use tables (SUTs) into a single supra-regional SUT. The method corrects for double-counting of intragroup re-exports, rebalances intra-group imports with the intra-group exports, and re-prices intra-group import matrices from c.i.f. prices to basic prices of the exporting industries. The method was tested with a seven-fold decomposition of the errors made in the estimation of the EU27 value added embodied in the EU27 exports to third countries for the years 2000-2007. With this new dataset, comparable estimates may now also be made for, inter alia, embodied CO_2 -emissions, water use and employment.²⁰

The first set of errors is due to the neglect of the intra-EU spillover and feedbacks effects that results if the domestic value added impacts, as calculated with the 27 national supply and use tables (SUTs), are simply summed. This leads to an underestimation of the EU27 value added embodied in the EU exports of 12-15%. The second set of errors relates to using the simple sum of 27 national SUTs instead of the consolidated EU27 with the above mentioned corrections included. Not using the correctly consolidated EU27 SUT leads to a further underestimation of the embodied value added of 11-16% over the period 2000-2007.

²⁰ See, e.g., Moll and Rémond-Tiedrez (2011) for a comparable CO₂ application.

Moreover, large industry-by-industry variations in errors of up to plus and minus 50% were observed behind these aggregate errors.

Even with these underestimations removed, the total EU27 value added embodied in the EU27 exports to third countries amounts to only a little over 11% of the gross domestic product (GDP) of the EU27. The remaining 89% is related to EU27 domestic private and public consumption and investment expenditures. This suggests that internal EU27 innovativeness and market efficiency might be more important for its GDP growth than the EU27's external competitiveness.

Acknowledgements

This article is a spin-off of two projects financed by Eurostat. We thank Stephan Moll, and in particular Jörg Beutel, for extensive discussions on the methodology. The views expressed in this article are solely the views of the authors and do not necessarily reflect those of Eurostat or the European Commission services.

References

- Bacharach, M. (1970) *Biproportional matrices & input-output change.* Cambridge University Press.
- Belke A. & L. Wang (2006) The degree of openness to intra-regional trade Towards valueadded based openness measures. *Jahrbücher für Nationalökonomie und Statistik*, 226, 115-138.
- Brahmbhatt, M. (1998) *Measuring Global Economic Integration A Review of the Literature and Recent Evidence*. World Bank, Washington D.C.
- Brockmeier M. (1997) *GTAP 3 Data Base Documentation Chapter 16.2 EU-12*. Center for Global Trade Analysis, Purdue University.
- Council of the European Union (2009) *Conclusions on "Priorities for the Internal Market in the next decade" contribution by the Competitiveness Council to the post 2010 Lisbon agenda*. 2982nd Competitiveness Council meeting, Brussels.
- Daudin G. C. Rifflard & D. Schweisguth (2011) Who produces for whom in the world economy? *Canadian Journal of Economics*, 44, 1403-1437.
- Eder P., L. Delgado, F. Neuwahl, A. Tukker, G. Huppes, L. van Oers & R. Heijungs (2006) Environmentally extended input-output tables and models for Europe. Report EUR

22194 EN, European Commission, Joint Research Center, Institute for Prospective Technological Studies, Seville.

- European Commission (2010a) *Trade as a driver of prosperity.* Commission staff working document accompanying the Commission's Communication on "Trade, Growth and World affairs.", SEC(2010)1269, DG Trade: Brussels.
- European Commission (2010b) *Trade, growth and world affairs. Trade policy as a core component of the EU's 2020 strategy.* Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions, COM(2010)612, DG Trade: Brussels.
- European Commission, International Monetary Fund, Organisation for Economic Co-operation and Development, United Nations & World Bank (2009) *System of National Accounts* 2008.
- Eurostat (2006) Statistics on the trading of goods user guide (Manual No. KS-BM-06-001)
- Eurostat (2008) Eurostat Manual of Supply, Use and Input-Output Tables (Manual No. KS-RA-07-013-EN-N).
- Eurostat (2011) Creating consolidated and aggregated EU27 Supply, Use and Input-Output Tables, adding environmental extensions (air emissions), and conducting Leontief-type modeling to approximate carbon and other 'footprint' of EU27 consumption for 2000 to 2006. Eurostat, Luxembourg.
- Dietzenbacher, E., B. Los, R. Stehrer, M. Timmer & G. de Vries (2013) The Construction of World Input-Output Tables in the Wiod Project. *Economic Systems Research*, 25, 71-98.
- Foster N., R. Stehrer & G. de Vries (2011) Trade in value added and factors: A comprehensive approach. Unpublished manuscript, WIOD project.
- Hewings, G.J.D. (1977) Evaluating the Possibilities for Exchanging Regional Input-Output Coefficients, *Environment and Planning A*, 9, 927-944.
- Ilkovic F., A. Dierx, V. Kovacs & N. Sousa (2007) Steps towards a deeper economic integration: the Internal Market in the 21st century, A contribution to the Single Market Review. European Commission, DG Economic and Financial Affairs: Brussels.
- Isard, W. (1951) Interregional and regional input-output analysis, a model of a space economy. *Review of Economics and Statistics*, *33*, 318-28.
- Johnson R.C. & G. Noguera (2012) Accounting for intermediates: Production sharing and trade in value added. *Journal of International Economics*, 86, 224-236.

- Junius, T., & J. Oosterhaven (2003) The solution of updating or regionalizing a matrix with both positive and negative entries. *Economic Systems Research*, 15, 87-87.
- Koopman R., W. Powers, Z. Wang & S. Wei (2010) *Give credit where credit is due: tracing value added in global production chains*. NBER working paper no. 16426.
- Lahr, M. & L. de Mesnard (2004) Biproportional techniques in input-output analysis: Table updating and structural analysis. *Economic Systems Research*, 16, 115-134.
- Leeuwen, M. van (2002) *GTAP 5 Data Base Documentation Chapter 11.M The European Union*. Center for Global Trade Analysis, Purdue University.
- Leeuwen, M. van & D. Verhoog (1999) *GTAP 4 Data Base Documentation Chapter 14/17 European Union.* Center for Global Trade Analysis, Purdue University.
- Lenzen, M., L. Pade & J. Munksgaard (2004) CO₂ Multipliers in Multi-region Input–Output Models. *Economic Systems Research*, 16, 391-412.
- Linden, J.A. van der (1999) Interdependence and Specialisation in the European Union: Intercountry Input-Output Analysis and Economic Integration. Ph.D, Faculty of Economics, University of Groningen.
- Linden, J. A. van der & J. Oosterhaven (1995). European community intercountry input-output relations: Construction method and main results for 1965-85. *Economic Systems Research*, 7, 249-249.
- Meng, B., Y. Zhang & S. Inomata (2013) Compilation and Applications of the IDE-JETRO's International Input-Output Tables, *Economic Systems Research*, 25, 122-142.
- Miller, R.E. & P.D. Blair (2009) *Input-output analysis: Foundations and extensions*. Cambridge University Press.
- Minguez, R., J. Oosterhaven & F. Escobedo (2009) Cell-Corrected RAS method (CRAS) for Updating or Regionalizing an Input-Output Matrix. *Journal of Regional Science* 49(2): 329–348.
- Moll, S. & I. Rémond-Tiedrez (2011) CO₂ emissions induced by EU's final use of products are estimated to be 9 tonnes per capita. *Statistics in focus 22/2011*, Eurostat, Luxembourg.
- Narayanan, B., Aguiar, A. & R. McDougall (eds) (2012) Global Trade, Assistance, and Production: The GTAP 8 Data Base. Center for Global Trade Analysis, Purdue University. Available online at: http://www.gtap.agecon.purdue.edu/databases/v8/ v8_doco.asp

- Oosterhaven, J. (1981) Interregional Input-Output Analysis and Dutch Regional Policy Problems. Gower, Aldershot, UK.
- Oosterhaven, J. (1984) A family of square and rectangular interregional input-output tables and models. *Regional Science and Urban Economics*, 14, 565-82.
- Oosterhaven, J. (1995) Changing specialization and interdependency of EC-economies 1959-1975. *Australasian Journal of Regional Studies*, 1, 51-64.
- Oosterhaven, J., D. Stelder, & S. Inomata (2008) Estimating international interindustry linkages: Non-survey simulations of the Asian-Pacific economy. *Economic Systems Research*, 20, 395-414.
- Oosterhaven, J. & F. Escobedo-Cardeñoso (2011) A new method to estimate input-output tables by means of structural lags, tested on Spanish regions. *Papers in Regional Science*, 60, 829-845.
- Puzzello L. (2011) A proportionality assumption and measurement biases in the factor content of trade, *Journal of International Economics*, 87, 105-111.
- Roca Zamora A. (2009) How is the internal market integration performing? Trade and foreign direct investment indicators for monitoring the state of the economic integration. European Commission, DG Internal Market and Services: Brussels.
- Rueda-Cantuche, J.M. (2011) The choice of type of Input-Output Table revisited: moving towards the use of supply-use tables in impact analysis, *Statistics and Operations Research*, *35*(1), 21-38.
- Rueda-Cantuche, J. M., J. Beutel, F. Neuwahl, I. Mongelli, & A. Loeschel (2009) A Symmetric Input–Output Table For EU27: Latest Progress. *Economic Systems Research*, 21, 59-79.
- Schilderinck, J.H.F. (1984) Interregional Structure of the European Community, Part II. Research Memorandum No. 165, Department of Economics, Catholic University of Brabant, Tilburg.
- Stone, R. (1961) *Input-Output and National Accounts*. Organization for European Economic Cooperation, Paris.
- Tukker, A., A. de Koning, R. Wood, T. Hawkins, S. Lutter, J. Acosta, J.M. Rueda Cantuche, M. Bouwmeester, J. Oosterhaven, T. Drosdowski and J. Kuenen (2013) Exiopol Development and Illustrative Analyses of a Detailed Global MR EE SUT/IOT. *Economic Systems Research*, 25, 50-70.
- Ungar, P. & J. Heuschling (1994) The Interlinking of Economic Branches in the European Union
 Consolidated Input-Output Table for 1991. *Rapid Reports, Economy and Finance*, 2, Eurostat, Luxembourg.

List of research reports

12001-HRM&OB: Veltrop, D.B., C.L.M. Hermes, T.J.B.M. Postma and J. de Haan, A Tale of Two Factions: Exploring the Relationship between Factional Faultlines and Conflict Management in Pension Fund Boards

12002-EEF: Angelini, V. and J.O. Mierau, Social and Economic Aspects of Childhood Health: Evidence from Western-Europe

12003-Other: Valkenhoef, G.H.M. van, T. Tervonen, E.O. de Brock and H. Hillege, Clinical trials information in drug development and regulation: existing systems and standards

12004-EEF: Toolsema, L.A. and M.A. Allers, Welfare financing: Grant allocation and efficiency

12005-EEF: Boonman, T.M., J.P.A.M. Jacobs and G.H. Kuper, The Global Financial Crisis and currency crises in Latin America

12006-EEF: Kuper, G.H. and E. Sterken, Participation and Performance at the London 2012 Olympics

12007-Other: Zhao, J., G.H.M. van Valkenhoef, E.O. de Brock and H. Hillege, ADDIS: an automated way to do network meta-analysis

12008-GEM: Hoorn, A.A.J. van, Individualism and the cultural roots of management practices

12009-EEF: Dungey, M., J.P.A.M. Jacobs, J. Tian and S. van Norden, On trend-cycle decomposition and data revision

12010-EEF: Jong-A-Pin, R., J-E. Sturm and J. de Haan, Using real-time data to test for political budget cycles

12011-EEF: Samarina, A., Monetary targeting and financial system characteristics: An empirical analysis

12012-EEF: Alessie, R., V. Angelini and P. van Santen, Pension wealth and household savings in Europe: Evidence from SHARELIFE

13001-EEF: Kuper, G.H. and M. Mulder, Cross-border infrastructure constraints, regulatory measures and economic integration of the Dutch – German gas market

13002-EEF: Klein Goldewijk, G.M. and J.P.A.M. Jacobs, The relation between stature and long bone length in the Roman Empire

13003-EEF: Mulder, M. and L. Schoonbeek, Decomposing changes in competition in the Dutch electricity market through the Residual Supply Index

13004-EEF: Kuper, G.H. and M. Mulder, Cross-border constraints, institutional changes and integration of the Dutch – German gas market

13005-EEF: Wiese, R., Do political or economic factors drive healthcare financing privatisations? Empirical evidence from OECD countries

13006-EEF: Elhorst, J.P., P. Heijnen, A. Samarina and J.P.A.M. Jacobs, State transfers at different moments in time: A spatial probit approach

13007-EEF: Mierau, J.O., The activity and lethality of militant groups: Ideology, capacity, and environment

13008-EEF: Dijkstra, P.T., M.A. Haan and M. Mulder, The effect of industry structure and yardstick design on strategic behavior with yardstick competition: an experimental study

13009-GEM: Hoorn, A.A.J. van, Values of financial services professionals and the global financial crisis as a crisis of ethics

13010-EEF: Boonman, T.M., Sovereign defaults, business cycles and economic growth in Latin America, 1870-2012

13011-EEF: He, X., J.P.A.M Jacobs, G.H. Kuper and J.E. Ligthart, On the impact of the global financial crisis on the euro area

13012-GEM: Hoorn, A.A.J. van, Generational shifts in managerial values and the coming of a global business culture

13013-EEF: Samarina, A. and J.E. Sturm, Factors leading to inflation targeting – The impact of adoption

13014-EEF: Allers, M.A. and E. Merkus, Soft budget constraint but no moral hazard? The Dutch local government bailout puzzle

13015-GEM: Hoorn, A.A.J. van, Trust and management: Explaining cross-national differences in work autonomy

13016-EEF: Boonman, T.M., J.P.A.M. Jacobs and G.H. Kuper, Sovereign debt crises in Latin America: A market pressure approach

13017-GEM: Oosterhaven, J., M.C. Bouwmeester and M. Nozaki, The impact of production and infrastructure shocks: A non-linear input-output programming approach, tested on an hypothetical economy

13018-EEF: Cavapozzi, D., W. Han and R. Miniaci, Alternative weighting structures for multidimensional poverty assessment

14001-OPERA: Germs, R. and N.D. van Foreest, Optimal control of production-inventory systems with constant and compound poisson demand

14002-EEF: Bao, T. and J. Duffy, Adaptive vs. eductive learning: Theory and evidence

14003-OPERA: Syntetos, A.A. and R.H. Teunter, On the calculation of safety stocks

14004-EEF: Bouwmeester, M.C., J. Oosterhaven and J.M. Rueda-Cantuche, Measuring the EU value added embodied in EU foreign exports by consolidating 27 national supply and use tables for 2000-2007

www.rug.nl/feb